Author's response to reviews

Title: Individual patient data systematic review and meta-analysis of ultrasonography of optic nerve sheath diameter for detecting raised intracranial pressure: protocol of the ONSD Research Group

Authors:

Julie Dubourg (julie.dubourg@gmx.fr)
Mahmoud Messerer (m.messerer@laposte.net)
Dimitrios Karakitsos (karakitsosdimitrios@gmail.com)
Venkatakrishna Rajajee (vrajajee@yahoo.com)
Erik Antonsen (eantonsen@partners.org)
Etienne Javouhey (etienne.javouhey@chu-lyon.fr)
Alessandro Cammarata (alecamma81@hotmail.it)
Michael Cotton (michael.cotton@chuv.ch)
Roy Thomas Daniel (roy.daniel@chuv.ch)
Carmelo Denaro (carmelodenaro1@tin.it)
Emmanuel Douzinas (edouzin@med.uoa.gr)
Clément Dubost (clement.dubost@hotmail.com)
Moncef Berhouma (berhouma.moncef@yahoo.fr)
Behrouz Kassai (bk@ucpl.univ-lyon1.fr)
Muriel Rabilloud (muriel.rabilloud@chu-lyon.fr)
Antonino Gullo (a.gullo@policlinico.unict.it)
Abderrhammane Hamlat (abderrhammane.hamlat@chu-rennes.fr)
Gregorios Kouraklis (gkouraklis@hotmail.com)
Giuseppe Mannanici (guiseppe.mannanici@gmail.com)
Keith Marill (kmarill@partners.org)
Sybille Merceron (sybille07@gmail.com)
John Poularas (poularas@hotmail.com)
Giuseppe Ristagno (gristag@gmail.com)
Vicki Noble (vnoble@partners.org)
Sachita Shah (sachitas@uw.edu)
Heidi Kimberly (hkimberly@partners.org)
Gianluca Cammarata (cammarata.g@libero.it)
Riccardo Moretti (riccardo.moretti77@gmail.com)
Thomas Geeraerts (geeraerts.t@chu-toulouse.fr)

Version: 3 Date: 13 June 2013

Author's response to reviews: see over
RESPONSES TO REVIEWER

⇒ First, we want to thank you for your decision.

Please find above a point-by-point response to reviewer’s comments.

I am pleased to let you know that following review of your manuscript, the handling editor is happy to editorially accept your manuscript for publication on the condition that you address her comment below:

"Can you clarify with authors why there is a ?REF? in the paragraph Rationale for an individual patient data meta-analysis?"

⇒ This is a mistake. We deleted “REF”.